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Table 1 The element content in purification residue
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Fa 173 19.17 0.01 29.74 0.43 0.01 0.03 0.05

TE 7 = BACH g/t
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Ag,S0, +2NaCl—2AgCl | +Na,S0,  (5)
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Fig.1 Pyrometallurgical process flow of

purification residue
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Fig.2 Nitric acid treatment process of

purification residue
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Fig.3 Sulfuric acid treatment process of

purification residue
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Fig.4 Effect of sulfuric acid concentration

on copper leaching rate in purification residue
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Fig.5 Effect of reaction temperature on copper

leaching rate in purification residue
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Fig.6 Effect of reaction time on copper

leaching rate in purification residue
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Fig.7 Effect of liquid-solid ratio on copper

leaching rate in purification residue
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Fig.8 Effect of salt addition amount on silver

content of reaction mixture from purification residue
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Table 2 Test results of different amount of

purified slag

T Ak Ag. BT Ag, B
i
" Cu ¥/ % Cu F4/% W
H/t
Ag Cu Ag Cu (mg-L™")
1.2 20. 12 26. 81 71.31 0.32 0.5
1.5 18.91 28.90 70.20 0.30 0.7
1.8 17. 65 29. 56 71.02 0.33 0.6
2.0 16.70 30.23 71.12 0.35 0.7
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Study on separation of copper and silver from purification residue of silver electrolyte
ZHAO Zhupeng, ZHANG Shanhui, CUI Jiayou, ZHANG Teng, HE Dongxiao
(Shandong Humon Smelting Co. , Lid. , Yantai 264109, China)

Abstract; The purification slag produced in the purification process of silver electrolyte has a high copper content.
There are the features of long copper removal period, low gold and silver direct yield in the traditional pyrotechnic
process and the generated soot containing gold and silver needs secondary treatment, which increases the treatment
cost for the treatment of purification slag. In this paper, sulfuric acid method was used to treat the purification
residue produced in the purification process of silver electrolyte. The results showed that the leaching rate of copper
could reach more than 99% and the silver content in the solution could be reduced to 0. 5 mg/L by sulfuric acid wet
process with sulfuric acid concentration of 6% , reaction temperature of 75 °C, reaction time of 2 h, liquid-solid
ratio of 5: 1, and salt dosage of 11 g. The separation of silver and copper was realized and the silver content in the
leaching residue could be stabilized at more than 70% and the copper content was less than 0.4% by the
production process application. Compared with the traditional treatment process, this process has the advantages of
simple operation and good economic and environmental benefits.

Key words: silver electrolyte; purification residue; copper; sulfuric acid ;silver;wet process



